Author's response to reviews

Title: Effectiveness of IT-based diabetes management interventions: exploring mixed findings in the literature

Authors:

Beth M Costa (beth.costa@deakin.edu.au)
Kristine J Fitzgerald (kristine.fitzgerald@deakin.edu.au)
Kay M Jones (kay.jones@med.monash.edu.au)
Trisha Dunning AM (trisha.dunning@barwonhealth.org.au)

Version: 3 Date: 4 November 2009

Author's response to reviews:

To the BioMed Central Editorial Production Team

Thank you for your email dated 21/10/2009. Please find attached updated submission to the journal BMC Family Practice titled 'Effectiveness of IT-based diabetes management Interventions: a review of the literature'. The manuscript is 24 pages long and includes three tables.

Regarding amendments:

• we have amended the title to read “Effectiveness of IT-based diabetes management interventions: a review of the literature’ rather than the suggested amendment because this is not a systematic review,
• updated references 18 and 36
• removed ‘table placement’ text
• page 4 – included [10] in citation brackets
• deleted additional fines
• checked the manuscript for typographical errors and final corrections

Sincerely
Dr Kay Jones